%% Antes de processar este arquivo LaTeX (LaTeX2e) deve
%% verificar que o arquivo TEMA.cls estah no mesmo
%% diretorio. O arquivo TEMA.cls pode ser obtido do
%% endereco www.sbmac.org.br/tema.

\documentclass{TEMA2013}
%
\usepackage[latin1]{inputenc}
\usepackage[english]{babel}
%\usepackage{epsfig}
%
\usepackage{graphicx}
\usepackage{sidecap}
\usepackage{wrapfig}
\usepackage[normalem]{ulem}
\usepackage[mathscr] {eucal}
\usepackage{amssymb,amsmath}
\usepackage{arcs} % for placing an arc over (\overarc) or under (\underarc) a piece of text. (The text may be up to three letters long.)
%
\newcommand{\vetor}{\vec}
\newcommand{\iu}{\ensuremath{ {\rm i } }}
\newcommand{\itGamma} { \mathit{\Gamma} }
\newcommand{\itDelta} { \mathit{\Delta} }
\newcommand{\itPhi} { \mathit{\Phi} }
\newcommand{\itLambda} { \mathit{\Lambda} }
\newcommand{\itOmega} { \mathit{\Omega} }
\newcommand{\itPi} { \mathit{\Pi} }
\newcommand{\itSigma} { \mathit{\Sigma} }
\newcommand{\itTheta} { \mathit{\Theta} }
\newcommand{\itXi} { \mathit{\Xi} }
\newcommand{\itPsi} { \mathit{\Psi} }
\newcommand{\Real}{\ensuremath{ {\,\mathrm{Re} {\kern 1pt}} }}
%
\begin{document}
\begin{thebibliography}{11}

\bibitem{Barton}
G. Barton, ``Elements of Green's Functions and Propagation: Potentials, Diffusion, and Waves'', Oxford University Press, 1989.

\bibitem{Butkov}
E. Butkov, ``Mathematical Physics'', Addison-Wesley Publishing Company, Reading, Massachusetts, 1973.

\bibitem{Courant-Hilbert}
R. Courant, D. Hilbert, ``Methods of Mathematical Physics'', Volume II,  Third Printing, Interscience Publishers/John Wiley \& Sons, New York, 1966.

\bibitem{DaviesTAM41}
B. Davies, ``Integral Transforms and Their Applications'', Texts in Applied Mathematics 41, Third Edition, Springer-Verlag, New York, 2002.

\bibitem{Duffy}
D.G. Duffy, ``Green's Functions with Applications'', Studies in Advanced Mathematics, Chapman \& Hall/CRC Press LLC, Boca Raton, Florida, 2001.

\bibitem{Economou}
E.N. Economou, ``Green's Functions in Quantum Physics'', Third Edition, Springer Series in
Solid-State Sciences 7, Springer-Verlag, Berlin, 2006.

%\bibitem{Hadamard}
%J. Hadamard, ``Le problème de Cauchy et les équations aux dérivées partielles linéaires %hyperboliques'', Hermann et Cie, Paris, 1932.

\bibitem{HildebrandCALCULUS}
F.B. Hildebrand, ``Advanced Calculus for Applications'', Second Edition, Prentice-Hall, Englewood Cliffs, 1976.

\bibitem{Hsu}
H.P. Hsu, ``Applied Fourier Analysis'', HBJ Publishers, San Diego, 1984.

\bibitem{MorseFeshbach}
P.M. Morse and H. Feshbach, ``Methods of Theoretical Physics'', McGraw-Hill Book Company, New York, 1953.

\bibitem{ToscanoCoutoRBEF2}
R. Toscano Couto, Green's functions for the wave, Helmholtz and Poisson equations
in a two-dimensional boundless domain, \textit{Rev. Bras. Ens. Fis.}, {\bf 35}, No. 1 (2013), 1304.

\bibitem{Watson}
G.N. Watson, ``A Treatise on the Theory of Bessel Functions'', Second Edition, Cambridge University Press, London, 1944.
\end{thebibliography}


%********************************************************
\title
    {New Methodologies for the Calculation of Green's Functions for Wave Problems in Two-Dimensional Unbounded Domains}

\author
    {R.T. COUTO\thanks{toscano@im.uff.br}\;,
     Departamento de Matemática Aplicada, UFF - Universidade Federal Fluminense, 24020-140 Niterói, RJ, Brasil.}

\criartitulo

\runningheads {Couto}{New Methodologies for the Calculation of Green's Functions}

\begin{abstract}
{\bf Abstract}. This work describes the application of new methodologies for the evaluation of the inverse Fourier transforms that yield Green's functions for both the wave and Helmholtz equations in the entire bidimensional domain.

{\bf Keywords}. Wave equation, two dimensions, Green's function.
\end{abstract}


%********************************************************
\newsec{Introduction}
Green's functions for wave problems, both time-dependent and stationary ones, governed by the wave and Helmholtz equations, respectively, in unbounded domains having one, two or three dimensions have well known expressions (cf. reference \cite{Barton}, sections 11.2 and 13.2.2). The Fourier transform serves well in their determination, but the evaluation of the inversion integral in two dimensions -- the case considered in this work -- is the most challenging.

For the Helmholtz equation, reference \cite{MorseFeshbach}, on pp.~822-824, states that the inversion can be performed by using contour integration together with a change of complex variables of the type given in equation (\ref{greenE70}) below, but does not reveal the steps of the calculation. Later on, reference \cite{DaviesTAM41}, on pp.~173-176, shows a little more thereof, but still as an outline which is hard to follow.

It is thus our purpose to offer here a detailed description of this methodology, but, to make an innovation, we apply it to the wave equation. Both retarded (Section 2) and advanced (Section 3) Green's functions are calculated. Green's functions for the Helmholtz equation are also obtained as a by-product (Section 4). We conclude the exposition with final comments (Section 5).
\newsec{Retarded Green's Function for the Wave Equation}
Green's function $G(\vetor{r},t\,|\,\vetor{r}\,',t')$ for the wave equation in a boundless two-dimensional domain is the solution of
\begin{equation}\label{greenE10}
\nabla^2 G(\vetor{r},t\,|\,\vetor{r}\,',t') - \frac{1}{c^2}\,\frac{\partial^2G}{\partial t^2} = \delta(\vetor{r}-\vetor{r}\,')\,\delta(t-t')\;\;,
\end{equation}
with $\vetor{r}$ and $\vetor{r}\,'$ in $\mathbb{R}^2$, and $t$ and $t'$ in $\mathbb{R}$. In this section, we consider the \emph{retarded} or \emph{causal} Green's function, for which
\begin{equation}\label{greenE20}
G(\vetor{r},t\,|\,\vetor{r}\,',t') = 0 \quad\textrm{if}\quad t < t'\;\;.
\end{equation}

To solve the problem defined by (\ref{greenE10}) and (\ref{greenE20}), we first take the Fourier transform of (\ref{greenE10}) with respect to $t$, obtaining
\begin{equation}\label{greenE30}
\nabla^2 \tilde G(\vetor{r},\omega\,|\,\vetor{r}\,',t') +(\omega/c)^2 \tilde G = \delta(\vetor{r}-\vetor{r}\,')\,e^{\iu\omega t'}\!/\sqrt{2\pi}\;\;,
\end{equation}
where
\[
\tilde G(\vetor{r},\omega\,|\,\vetor{r}\,',t') \equiv \mathscr{F}_t\{G(\vetor{r},t\,|\,\vetor{r}\,',t')\}=\frac{1}{\sqrt{2\pi}} \int_{-\infty}^{\:\infty}\!dt\, e^{\iu\omega t} G(\vetor{r},t\,|\,\vetor{r}\,',t')\;\;.
\]
To compute the inverse transform $G = \mathscr{F}^{\,-1}_t\{\tilde G\}=(2\pi)^{-1/2}\!\int_{-\infty}^{\infty}d\omega\,e^{-\iu\omega t}\tilde G$, we modify this formula a little, by splitting the integral in the intervals $(-\infty,0)$ and $(0,\infty)$ and performing the changing of variable $\omega\rightarrow-\omega$ in the first integral, obtaining
\begin{equation}\label{greenE35}
G(\vetor{r},t\,|\,\vetor{r}\,',t') = \frac{1}{\sqrt{2\pi}} \int_0^{\:\infty}\!d\omega \big[ e^{-\iu\omega t} \tilde G(\vetor{r},\omega\,|\,\vetor{r}\,',t') + e^{\iu\omega t} \tilde G(\vetor{r},-\omega\,|\,\vetor{r}\,',t')\,\big] \;\;.
\end{equation}
By using this formula, we avoid negative values of $\omega$, what simplifies the development of the method.

Next, adopting the Cartesian coordinates $x$ and $y$ of $\vetor{r}$, in terms of which $\nabla^2\tilde G=\partial^2\tilde G/\partial x^2 + \partial^2\tilde G/\partial y^2$ and $\delta(\vetor{r}-\vetor{r}\,')=\delta(x-x')\delta(y-y')$, we take another Fourier transform, now with respect to $y$, obtaining
\begin{equation}\label{greenE40}
\frac{d^2\bar{\tilde G}}{dx^2}(x,k,\omega\,|\,x',y',t')-\left(k^2-\frac{\omega^2}{c^2}\right)\bar{\tilde G}=\frac{e^{\iu(ky'+\omega t')}}{2\pi}\,\delta(x-x')\;\;,
\end{equation}
where
\[
\bar{\tilde G}(x,k,\omega\,|\,x',y',t') \equiv \mathscr{F}_y\{\tilde G(x,y,\omega\,|\,x',y',t')\}=\!\frac{1}{\sqrt{2\pi}}\!\! \int\limits_{-\infty}^{\:\infty}\!\!dy\, e^{\iu ky} \tilde G(x,y,\omega\,|\,x',y',t')\;.
\]

We then solve the ordinary differential equation (\ref{greenE40}) under the conditions of continuity and finiteness for all $x$ as well as an extra condition (as a consequence of the delta function) which follows from its integration in the infinitesimal interval $(x'-\varepsilon,\:x'+\varepsilon)\;$ \{cf. \cite{Butkov}, section~12.2\}:
\[
\int_{-\varepsilon}^{\varepsilon}dx\,\frac{d^2\bar{\tilde G}}{dx^2}(x,k,\omega\,|\,x',y',t') - \left(k^2-\frac{\omega^2}{c^2}\right)\int_{-\varepsilon}^{\varepsilon}dx\,\bar{\tilde G}=\frac{e^{\iu(ky'+\omega t')}}{2\pi}\int_{-\varepsilon}^{\varepsilon}dx\,\delta(x-x')\;\;.
\]
The second integral above tends to zero, because it is the integral of a continuous function in a infinitesimal interval, and the last integral is equal to one. Carrying out the first integral and letting $\varepsilon\to0^+$, we obtain the \emph{jump condition} for $d \bar{\tilde G}/dx$ at $x=x'$:
\begin{equation}\label{greenE50}
\frac{d\bar{\tilde G}}{dx}(x'^+,k,\omega\,|\,x',y',t') - \frac{d\bar{\tilde G}}{dx}(x'^-,k,\omega\,|\,x',y',t') = \frac{e^{\iu(ky'+\omega t')}}{2\pi} \;\;.
\end{equation}

Notice that (\ref{greenE40}) is a homogeneous differential equation, except for $x=x'$; its solution for $k\ne\omega/c$ is thus of the form
\begin{equation}\label{greenE54}
\left. \bar{\tilde G}(x,k,\omega\,|\,x',y',t') \right|_{k \ne \frac{\omega}{c}} =\left\{\!\!
\begin{array}{lcl}
c_1\,e^{(a+b\iu)x}+c_2\,e^{-(a+b\iu)x} & (x<x') & \\
d_1\,e^{(a+b\iu)x}+d_2\,e^{-(a+b\iu)x} &  (x>x') &.
\end{array}
\right.
\end{equation}

In this equation, $\sqrt{k^2-\omega^2/c^2}\equiv \pm(a+b\iu)$ (decomposition of the square roots in their real and imaginary parts). Also, because of (\ref{greenE50}), it was necessary to consider arbitrary constants for $x<x'$, $c_1$ and $c_2$, different from those for $x>x'$, $d_1$ and $d_2$. These constants are to be determined by imposing the finiteness, continuity and jump conditions. Once found $\bar{\tilde G}(x,k,\omega\,|\,x',y',t')$, we can begin calculating the inverse Fourier transforms, $\mathscr{F}^{\,-1}_y\{\bar{\tilde G}\}=\tilde G$ first:

\begin{equation}\label{greenE57}
\tilde G(x,y,\omega\,|\,x',y',t') = \frac{1}{\sqrt{2\pi}} \int_{-\infty}^{\infty} dk\; e^{-\iu k y}\,\bar{\tilde G}(x,k,\omega\,|\,x',y',t')\;\;.
\end{equation}

But before doing so, let us make three observations:

\begin{enumerate}
\item For \emph{real} $k$ and $\omega$ (variables introduced in the Fourier transforms), either $a$ or $b$ vanishes, that is, $ab=0$. But, in this work, $k$ is not a real variable. In fact, the method described herein consists in evaluating the Fourier inversion integral in (\ref{greenE57}) by considering it as a contour integral along the real axis of the $k$-plane and then deforming this path of integration into another like those in Figures \ref{F3} to \ref{F5}. Clearly, for the complex values of $k$ along the new paths, $ab\ne0$ most often.
\item We need to consider only $a>0$. In fact, that we can take $a\ge0$ without loss of generality is obvious, and the results for $a=0$ are not necessary for the following reason: The set $S$ of points of the $k$-plane corresponding to $a=0$ are those in the real axis segment between $-\omega/c$ and $\omega/c$ as well as in the imaginary axis, and Figures \ref{F3} to \ref{F5} show that any of the new paths of integration contains a finite number of points of $S$ (two points, to be precise). This means that a finite number of values of $\bar{\tilde G}(x,k,\omega\,|\,x',y',t')$ given by (\ref{greenE54}) with $a=0$ are integrated along the new paths [in contrast with the infinite number of values along the real path in (\ref{greenE57})], and since these values are finite, their contribution to the integral is negligible. As a conclusion, there is no need to consider $a=0$.
\item Notice that (\ref{greenE54}) is not valid for $k=\omega/c$ (that is, for $a=b=0$); but a valid expression for this case is not necessary, because the point $k=\omega/c$ never belongs to the new path of integration (c.f. Figures \ref{F3} to \ref{F5}).
\end{enumerate}

Let us now proceed completing the determination of $\bar{\tilde G}$. In (\ref{greenE54}), we set $c_2=d_1=0$ to avoid infinite values for $x\to\pm\infty$. Requiring continuity at $x=x'$, that is, $G(x'^+,k,\omega\,|\,x',y',t')=G(x'^-,k,\omega\,|\,x',y',t')$, we can eliminate $d_2$, obtaining

\[
\bar{\tilde G}(x,k,\omega\,|\,x',y',t')=\left\{\!\!
\begin{array}{lcl}
c_1\:e^{(a+b\iu)x} & (x\le x') & \\
c_1\:e^{2(a+b\iu)x'}e^{-(a+b\iu)x} &  (x\ge x') &.
\end{array}
\right.
\]

By using (\ref{greenE50}), the jump condition, we find
$c_1=-\frac{e^{-(a+b\iu)x'}e^{\iu(ky'+\omega t')}}{4\pi(a+b\iu)}\,$,
whose substitution in the equation above furnishes the desired solution of (\ref{greenE40}):

\[
\bar{\tilde G}(x,k,\omega\,|\,x',y',t')=
-\frac{e^{\iu(ky'+\omega t')}}{4\pi(a+b\iu)}\!\times\!\!\left\{\!\!
\begin{array}{lcl}
e^{-(a+b\iu)(x'-x)} & (x\le x') & \\
e^{-(a+b\iu)(x-x')} &  (x\ge x') &,
\end{array}
\right.
\]
or
\[
\bar{\tilde G}(x,k,\omega\,|\,x',y',t')=-\frac{e^{
-\sqrt{k^2-\omega^2/c^2}\:|x'-x|\:+\:
\iu(ky'+\omega t')}}{4\pi\sqrt{k^2-\omega^2/c^2}} \;\;,
\]

with $\;\Real\sqrt{k^2-\omega^2/c^2}\,>0\;$ (because $a>0$).

We now use this result in the inversion Fourier integral given by (\ref{greenE57}):

\begin{equation}\label{greenE60}
\tilde G(x,y,\omega\,|\,x',y',t')=\frac{-1}{4\pi\sqrt{2\pi}} \int\limits_{-\infty}^{\infty} \frac{dk}{\sqrt{k^2-\omega^2/c^2}} \: e^{
-|X|\sqrt{k^2-\omega^2/c^2}\:-\:\iu(kY-\omega t')} ,
\end{equation}

where $X\equiv x-x' $ and $Y\equiv y-y' $.

Considering, as already mentioned, the integral in (\ref{greenE60}) as a contour integral along the real axis of the complex plane of $k=k_x+\iu k_y$, let us change the variable $k$ to another complex variable $\zeta = \phi+\iu u$ as follows\footnote{Instead of (\ref{greenE70}), we could have performed the change of variables $k=(\omega/c)\cos(\phi+\iu u)$, or even $k=(\omega/c)\sin(\phi+\iu u)$ with $\phi\in[-\pi/2,\,\pi/2]$ and $u\in(-\infty,\infty)$, and only a few modifications in the development would be necessary.}:
\begin{equation}\label{greenE70}
k=-\frac{\omega}{c}\cos\zeta=-\frac{\omega}{c}\cos(\phi+\iu u) = \underbrace{-\frac{\omega}{c}\cos\phi\cosh u}_{k_x} \,+ \;\iu \underbrace{\frac{\omega}{c}\sin\phi\sinh u}_{k_y}\;\;,
\end{equation}
from which, as indicated,
\begin{equation}\label{greenE80}
k_x = -(\omega/c)\cos\phi\cosh u \quad \textrm{and} \quad k_y = (\omega/c)\sin\phi\sinh u \;\;.
\end{equation}
\begin{wrapfigure}[11]{r}{6.0cm}\centering
\includegraphics{fig01.eps}
\caption{The domain and image of the map defined by (\ref{greenE70}).}
\label{F1}
\end{wrapfigure}
These equations with $\phi\in[0,\pi]$ and $u\in(-\infty,\infty)$ define a map from the strip of the $\zeta$-plane shown in Figure \ref{F1} to the whole $k$-plane. Figure \ref{F2} shows that a vertical straight line $\phi=$ constant ($\ne0,\pi/2$ or $\pi$) is mapped to a hyperbola (in the left half plane if $\phi=\phi_1<\pi/2$ or the right one if $\phi=\phi_2>\pi/2$) and that a horizontal line segment $u=$ constant $(\ne0)$ is mapped to a half ellipse (in the upper half plane if $u=u_1>0$ or the lower one if $u=u_2<0$). In fact, in the $k$-plane, (\ref{greenE80}) with $\phi=\phi_0\;(\ne0,\pi/2$ or $\pi)$ or $u=u_0 \;(\ne0)$ can be seen respectively as the parametrization of:
\begin{itemize}
\item the left (if $\phi_0<\pi/2$) or right (if $\phi_0>\pi/2$) branch of the hyperbola
    \[
    \left[\frac{k_x}{-(\omega/c)\cos\phi_0}\right]^2- \left[\frac{k_y}{(\omega/c)\sin\phi_0}\right]^2=\:1
    \]
\item the upper (if $u_0>0$) or lower (if $u_0\!<0$) half of the ellipse
    \[
    \left[\frac{k_x}{-(\omega/c)\cosh u_0}\right]^2 + \left[\frac{k_y}{(\omega/c)\sinh u_0}\right]^2=\:1
    \]
\end{itemize}
In addition, (\ref{greenE80}) with $\phi=0$ or $\pi$ describes the portion of the real axis from $-\infty$ to $-(\omega/c)$ or that from $(\omega/c)$ to $\infty$, respectively; with $\phi=\pi/2$, the imaginary axis; and with $u=0$, the portion of the real axis from $-\omega/c$ to $\omega/c$.
\begin{figure}[h!]\centering
\includegraphics[width=\linewidth]{fig02.eps}
\caption{The map in (\ref{greenE70}): hyperbolas and half ellipses as images of vertical straight lines and horizontal line segments, respectively. (A line in the $\phi u$-plane and its image are both drawn with the same pattern and oriented with the same kind of arrow.)}
\label{F2}
\end{figure}
\begin{figure}[h!]
\centering
\includegraphics{fig03.eps}
\caption{The contour $C$ used to evaluate the integral in (\ref{greenE60}).}
\label{F3}
\end{figure}

We will evaluate the integral in (\ref{greenE60}) for the contour $C = E_1\cup H\cup E_2$ depicted in Figure \ref{F3}, but with $R\to\infty$, where:
\begin{itemize}
\item $E_1$ is the elliptical path given by (\ref{greenE80}) with $u=u_0 = \cosh^{-1}(cR/\omega)$ and $\phi$ varying from 0 to a suitable value $\phi_0$ yet to be determined
\item $H$ is the hyperbolic path given by (\ref{greenE80}) with $\phi=\phi_0$ and $u$ varying from $u_0$ to $-u_0$
\item $E_2$ is the elliptical path given by (\ref{greenE80}) with $u=-u_0$ and $\phi$ varying from $\phi_0$ to $\pi$
\end{itemize}

In (\ref{greenE60}) we are faced with the problem of choosing the correct branch of the square root $z(k)\equiv\sqrt{k^2-\omega^2/c^2}=\pm|z(k)|\exp[\iu\arg z(k)]$. Let us proceed considering both branches simultaneously (we will reach a point at which consistency will impose the correct one):
\[
\sqrt{k^2-\omega^2/c^2}=\pm\iu\,(\omega/c)\sin\zeta=\pm\iu\,(\omega/c)\sin(\phi+\iu u)\;\;.
\]

Therefore, using this and (\ref{greenE70}), we have that
\begin{equation}\label{greenE90}
\frac{dk}{\sqrt{k^2-\omega^2/c^2}}=\frac{(\omega/c)\sin\zeta\,d\zeta} {\pm\iu\,(\omega/c)\sin\zeta}=\mp\iu\,d\zeta=\mp\iu\,(d\phi+\iu du)=\mp(-du+\iu d\phi)
\end{equation}

and also that the exponent appearing in (\ref{greenE60}) can be written as follows:
\[
-|X|\sqrt{k^2-\omega^2/c^2}\:-\:\iu\,(kY-\omega t') =
\]
\[
-|X|\Big[\pm\iu\, \frac{\omega}{c}\sin(\phi+\iu u)\Big] \:-\:\iu\,\Big[-\frac{\omega}{c} \cos(\phi+\iu u)Y-\omega t'\Big] =
\]
\[
\mp\,\iu\,\frac{\omega}{c}|X|\,(\sin\phi\cosh u+\iu\cos\phi\sinh u)\:-\:\iu\,\Big[-\frac{\omega}{c}(\cos\phi\cosh u-\iu\sin\phi\sinh u)Y-\omega t'\Big]
\]
\[
=\frac{\omega}{c}\big(\pm|X|\cos\phi+Y\sin\phi \big)\sinh u+\iu\,\frac{\omega}{c}\big[(\mp|X|\sin\phi+Y\cos\phi)\cosh u+ct'\big]\;\;,
\]
or
\[
f(\phi,u) \equiv -|X|\sqrt{k^2-\omega^2/c^2}\:-\:\iu\,(kY-\omega t') =
\]
\begin{equation}\label{greenE100}
\frac{\omega}{c}g(\phi)\sinh u + \iu\frac{\omega}{c}\big[g'(\phi)\cosh u + ct'\big] \;\;,
\end{equation}
where
\begin{equation}\label{greenE110}
g(\phi) \equiv \pm|X|\cos\phi+Y\sin\phi \;\;.
\end{equation}

The integral in (\ref{greenE60}) for the contour $C$ can be split in three integrals evaluated on the three paths $E_1$, $H$ and $E_2$ which compose $C$. Thus, using (\ref{greenE90}) and (\ref{greenE100}), we can write (\ref{greenE60}) as follows:
\begin{eqnarray}
&\displaystyle \tilde G(\vetor{r},\omega\,|\,\vetor{r}\,',t') =
\frac{-1}{4\pi\sqrt{2\pi}}\left[ \int_{E_1}+\int_{H}+\int_{E_2}\right]e^{f(\phi,u)}\:
\underbrace{
dk/\sqrt{k^2\!-\omega^2/c^2}
}_{
\mp(-du+\iu\,d\phi)
}
&\nonumber \\
&\displaystyle = \frac{\pm 1}{4\pi\sqrt{2\pi}}\bigg[ \,\iu\int_0^{\phi_0}d\phi\:e^{f(\phi,u_0)}-\int_{u_0}^{-u_0}du\,e^{f(\phi_0,u)}+ \iu\int_{\phi_0}^{\pi}\:d\phi\:e^{f(\phi,-u_0)}\bigg]\,.\quad &  \label{greenE120}
\end{eqnarray}

Now is the moment to determine $\phi_0$. This parameter can be found in such a way that the two integrals with respect to $\phi$ in (\ref{greenE120}) (those evaluated on the elliptical part of $C$) go to zero as $R = (\omega/c)\cosh u_0 \to\infty$ (that is $u_0\to\infty$), thereby  considerably simplifying the calculations. Indeed, using (\ref{greenE100}), we see that these two integrals will tend to zero as $u_0\to\infty$ if $\Real{f(\phi,\pm u_0)} = \pm(\omega/c)\,g(\phi)\sinh u_0\to-\infty$, what will occur if $g(\phi)<0$ in the first integral with respect to $\phi$ and $g(\phi)>0$ in the second one. This imposes the requirements
\begin{equation}\label{greenE130}
\textrm{(a)}\;\;\; g(\phi_0)=0 \qquad\;\;\textrm{and} \qquad\;\; \textrm{(b)}\;\;\;g'(\phi_0)>0\;\;.
\end{equation}

Looking at (\ref{greenE120}) and (\ref{greenE100}), we see that we actually do not need $\phi_0$, but $g(\phi_0)$ and $g'(\phi_0)$. In order to calculate $g'(\phi_0)$, we need to develop (\ref{greenE130}a). Considering (\ref{greenE110}), we have
\begin{eqnarray*}
&g(\phi_0) = \pm |X|\cos\phi_0+Y\sin\phi_0 = 0 \;\;\Rightarrow\;\; X^2\cos^2\phi_0= Y^2\sin^2\phi_0&\\
&\Rightarrow\;\; X^2(1-\sin^2\phi_0) = Y^2\sin^2\phi_0\;\;\Rightarrow\;\; X^2=(X^2+Y^2)\sin^2\phi_0 \;\;,&
\end{eqnarray*}
from which, solving for $\sin\phi_0$ [the positive value is taken, because $\phi_0\in(0,\pi)$] and then calculating $\cos\phi_0$, we obtain
\[
\sin\phi_0 = |X|/\rho \qquad \textrm{and} \qquad \cos\phi_0 = \mp Y\sin\phi_0/|X| = \mp Y/\rho \;\;,
\]
where $\rho = \sqrt{X^2+Y^2} = \sqrt{(x-x')^2+(y-y')^2} = |\vetor{r}-\vetor{r}\,'| $ .\\
\begin{wrapfigure}[6]{l}{6.0cm}
\caption{The contour $C$ for $\cos\phi_0<0$.}
\includegraphics{fig04.eps}
\label{F4}
\end{wrapfigure}
(Notice that $\cos\phi_0$ may be negative, that is, $\phi_0>\pi/2\,$; in this case, the contour $C$ is like that in Figure $4$.)  Therefore, using these results, we get
\begin{eqnarray}
&g'(\phi_0) = \mp |X|\sin\phi_0+Y\cos\phi_0& \nonumber\\
&=\mp X^2\!/\rho \mp Y^2\!/\rho& \nonumber\\
&= \mp (X^2\!+Y^2)/\rho = \mp\rho \;\;.\;\;\;& \label{greenE140}
\end{eqnarray}

In this, in view of (\ref{greenE130}b), we choose the plus sign. Since this sign is the lower one in the ``$\mp$'' appearing in (\ref{greenE140}), in each ``$\pm$'' and ``$\mp$'' related to the two branches of $\sqrt{k^2-\omega^2/c^2}$, the lower sign is also the correct one.
The substitution of (\ref{greenE130}a) and $g'(\phi_0)=\rho$ in (\ref{greenE100}) then yields
\begin{equation}\label{greenE150}
f(\phi_0,u) = \frac{\omega}{c}\underbrace{g(\phi_0)}_{0}\sinh u + \iu\frac{\omega}{c}\Big[\underbrace{g'(\phi_0)}_{\rho}\cosh u + ct'\Big] = \iu\omega \left[\frac{\rho}{c}\cosh u + t'\right] \;\;.
\end{equation}

With this and the fact that the first and third integrals tend to zero, we can rewrite (\ref{greenE120}) as
\begin{equation}\label{greenE160}
\tilde G(\vetor{r},\omega\,|\,\vetor{r}\,',t') =
 \frac{-1}{4\pi\sqrt{2\pi}} \int_{-\infty}^{\infty}du\,e^{\iu\omega \left[\frac{\rho}{c}\cosh u + t'\right]} \;\;.
\end{equation}

Using (\ref{greenE35}) to calculate $\mathscr{F}^{\,-1}_t$ of this result, we obtain
\begin{eqnarray*}
G(\vetor{r},t\,|\,\vetor{r}\,',t') &=& \frac{-1}{8\pi^2} \int_{-\infty}^{\infty}du
\int_0^{\infty}d\omega \left\{
e^{ \iu\omega \left[\frac{\rho}{c}\cosh u -(t-t')\right]}+
e^{-\iu\omega \left[\frac{\rho}{c}\cosh u -(t-t')\right]}
\right\}\\
&=& \frac{-1}{4\pi} \int_{-\infty}^{\infty}du
\;\left\{\frac{1}{\pi}\int_0^{\infty}d\omega\:
\cos\omega\!\left[\frac{\rho}{c}\cosh u -(t-t')\right]
\right\}\;\;.
\end{eqnarray*}

Defining $T\equiv t-t'$, recognizing \{cf. \cite{Hsu}, equation~(6.28)\} that the last pair of braces encloses an integral representation of the delta function
\[
\delta\,[\,(\rho/c)\cosh u - T\,] = \delta\,[\,(\rho/c)\,(\cosh u - cT/\rho)\,] = (c/\rho)\,\delta\,(\cosh u - cT/\rho)\;\;,
\]
and changing to the variable $v=\cosh u$, we can proceed the calculation as follows:
\begin{eqnarray}
G(\vetor{r},t\,|\,\vetor{r}\,',t') &=& \frac{-1}{4\pi}\:2\!\int\limits_{0}^{\infty}du \: \frac{c}{\rho}\:\delta\Big(\cosh u - \frac{cT}{\rho}\Big) = \frac{-c}{2\pi\rho}\int\limits_1^\infty\frac{dv}{\sqrt{v^2-1}} \:\delta\Big(v-\frac{cT}{\rho}\Big) \phantom{ \frac{}{\frac{}{\int}} }\nonumber\\
&=& \frac{-c}{2\pi\rho\sqrt{(cT/\rho)^2-1}}\times
\underbrace{
\left\{\!\!
\begin{array}{lllll}
0 & \textrm{if} & cT/\rho<1 & \textrm{i.e.}& -\rho/c+T<0 \\
1 &  \textrm{if} & cT/\rho>1 & \textrm{i.e.}& -\rho/c+T>0
\end{array}\right.
}_{\mathcal{U}\,(-\rho/c+T)}{}^{}_{\displaystyle,}\nonumber
\end{eqnarray}
where $\mathcal{U}(\tau)$ is the unit step function (equal to 0 for $\tau<0$ and to 1 for $\tau>0$). We thus obtain the final result
\begin{equation}\label{greenE170}
G(\vetor{r},t\,|\,\vetor{r}\,',t') = G(\rho,T) = \frac{-1}{2\pi} \,\frac{\mathcal{U}(-\rho/c+T)}{\sqrt{-(\rho/c)^2+T^{\,2}}}\quad\left[\,\rho\equiv |\,\vetor{r}-\vetor{r}\,'|\,,\;\:T\equiv t-t'\,\right]\;\;,
\end{equation}
which, except for the multiplicative constant (due to little differences in the form of the wave equation considered), is the same expression obtained in references \cite[equation~(11.2.21)]{Barton} and \cite[p.~842, equation~(7.3.15)]{MorseFeshbach}, by using another method (by integrating the corresponding three-dimensional Green's function).
\newsec{Advanced Green's Function for the Wave Equation}
In the previous section, it looks like (\ref{greenE20}) is never used. Nevertheless, the Green's function given by (\ref{greenE170}) indeed satisfies that causality condition:
\[
t<t'\;\Rightarrow\;-\frac{\rho}{c}\,+\,T=-\frac{\rho}{c}\,+\,\underbrace{t-t'}_{<0}<0 \;\Rightarrow\;\mathcal{U}(\underbrace{-\rho/c+T}_{<0})=0\;\Rightarrow\; G(\vetor{r},t\,|\,\vetor{r}\,',t')=0\;.
\]

We then may ask: To obtain the advanced Green's function, satisfying
\begin{equation}\label{greenE180}
G(\vetor{r},t\,|\,\vetor{r}\,',t') = 0 \quad\textrm{if}\quad t > t'\;\;,
\end{equation}
what should be modified in the calculational method described above? The answer is simple but subtle: it is the contour used to perform the inversion integral (\ref{greenE60}) that needs modification. Incidentally, a contour formed with the hyperbolic and elliptical arcs that arise in the change of variable given by (\ref{greenE70}) is either compatible with (\ref{greenE20}) or (\ref{greenE180}).  The contour in Figure \ref{F3} is compatible with (\ref{greenE20}). If we want (\ref{greenE180}) to be satisfied, the contour $C$ to be used is that in Figure \ref{F5}. Let us confirm this.
\begin{figure}[h!]
\centering
\includegraphics{fig05.eps}
\caption{The contour along which the integral in (\ref{greenE60}) leads to the advanced Green's function.}
\label{F5}
\end{figure}

The integral in (\ref{greenE60}) for the contour in Figure \ref{F5} can be written in the form of (\ref{greenE120}) with a few obvious changes:

\begin{equation}\label{greenE190}
\tilde G(\vetor{r},\omega\,|\,\vetor{r}\,',t') =
\frac{\pm 1}{4\pi\sqrt{2\pi}}\Bigg[ \iu\int\limits_0^{\phi_0}\!d\phi\:e^{f(\phi,-u_0)}-\int\limits_{-u_0}^{u_0}\!\!du \,e^{f(\phi_0,u)}+ \iu\int\limits_{\phi_0}^{\pi}\!d\phi\:e^{f(\phi,u_0)}\Bigg] ,
\end{equation}
where $f(\phi,u)$ is still given by (\ref{greenE100}) and (\ref{greenE110}), remembering that the ``$\pm$'' indicates the use of both branches of $\sqrt{k^2-\omega^2/c^2}$ . As $R\to\infty$ (or $u_0\to\infty$), $\Real{f(\phi,\mp u_0)}=\mp(\omega/c)\,g(\phi)\sinh u_0\to-\infty$ if $g(\phi)>0$ in the first integral above and $g(\phi)<0$ in the third one, meaning that $g(\phi_0)=0$, as in the previous case, but now $g'(\phi_0)<0$.
In this case, it is the upper sign in the ``$\mp$'' in (\ref{greenE140}) the correct one. Let us then proceed from (\ref{greenE190}) with: (a) the upper sign in ``$\pm$''; (b) $u_0\to\infty$, thereby making the first and third integrals go to zero; and (c) $f(\phi_0)=\iu\omega[(-\rho/c)\cosh u+t']$ (from (\ref{greenE150}) with $-\rho$ in place of $\rho$). By doing this, we obtain (\ref{greenE160}), but with $-\rho$ in place of $\rho$, which, developed as before, leads to a result similar to (\ref{greenE170}):
\[
G(\vetor{r},t\,|\,\vetor{r}\,',t') = \frac{-c}{2\pi\rho\sqrt{(cT/\rho)^2-1}}\times
\underbrace{
\left\{\!\!
\begin{array}{lllll}
0 & \textrm{if} & cT/(-\rho)<1 & \textrm{i.e.}&  -\rho/c-T<0 \\
1 &  \textrm{if} & cT/(-\rho)>1 & \textrm{i.e.}& -\rho/c-T>0
\end{array}\right.
}_{\mathcal{U}\,(-\rho/c-T)}{}^{}_{\displaystyle}
\]
\[
\Rightarrow\quad G(\vetor{r},t\,|\,\vetor{r}\,',t') = G(\rho,T) = \frac{-1}{2\pi} \,\frac{\mathcal{U}(-\rho/c-T)}{\sqrt{-(\rho/c)^2+T^{\,2}}}\;\;.
\]

This is the advanced Green's function, satisfying (\ref{greenE180}):
\[
t>t'\;\Rightarrow\;-\frac{\rho}{c}-T=-\frac{\rho}{c}-\underbrace{(t-t')}_{>0}<0 \;\Rightarrow\;\mathcal{U}(\underbrace{-\rho/c-T}_{<0})=0\;\Rightarrow\; G(\vetor{r},t\,|\,\vetor{r}\,',t')=0\;.
\]
\newsec{Green's Function for the Helmholtz Equation}

Green's function for the Helmholtz equation can be easily obtained from the previous results. In fact, with the definitions of the new constants $K\equiv\omega/c $ and $\alpha\equiv e^{\iu\omega t'}/\sqrt{2\pi}$ (in the present context, the parameters $t'$ and $\omega$ are irrelevant) as well as $\itGamma(\vetor{r}\,|\,\vetor{r}\,'\,)\equiv \tilde G(\vetor{r},\omega\,|\,\vetor{r}\,',t')$, (\ref{greenE30}) takes the common form of the equation whose solution is the Green's function $\itGamma(\vetor{r}\,|\,\vetor{r}\,'\,)$  for the Helmholtz equation:
\begin{equation}\label{greenE200}
\nabla^2\itGamma(\vetor{r}\,|\,\vetor{r}\,'\,)+K^2\itGamma =\alpha\,\delta(\vetor{r}-\vetor{r}\,')\;\;.
\end{equation}

Moreover, we have seen that (\ref{greenE160}), as it stands or with $-\rho$ in place of $\rho$, provides an integral representation for $\tilde G=\itGamma$. But these two forms, except for multiplicative constants, can be recognized as known integral representations for the first or the second Hankel function of order zero (cf. equations (10) and (11) in \cite{Watson}, \S6.21). We thus see that (\ref{greenE200}) has the two well-known elementary solutions
\[
\itGamma(\vetor{r}\,|\,\vetor{r}\,'\,)=\frac{-\alpha}{4\pi}\int_{-\infty}^{\infty}du\,e^{\pm\iu K\rho\cosh u}=\left\{\!\!
\begin{array}{ll}
(-\alpha\iu/4)\,H_0^{(1)}(K\rho) & \textrm{for the ``$+$'' sign}\\
\phantom{-}(\alpha\iu/4)\,H_0^{(2)}(K\rho) & \textrm{for the ``$-$'' sign}\;\;.
\end{array}\right.
\]

The decision to use either one, or a linear combination of the two, depends on whether the physical problem at hand involves only outgoing or incoming waves (cf. \cite{HildebrandCALCULUS}, Sec.~9.12, p.~470), or a superposition of these two kinds of waves.

\newsec{Conclusion}
The main part of the calculations developed above consists in solving Helmholtz equation (\ref{greenE30}) to obtain its solution in the form of the integral representation given by (\ref{greenE160}). This is actually the derivation of Green's function for Helmholtz equation (4.1), as explained in section 3 (where that integral representation is recognized as the first Hankel function of order zero). In the literature, there exists two alternative methods for the calculation of this Green's function by \emph{directly} solving the two-dimensional Helmholtz equation as well as two \emph{indirect} methods, in which that equation is not solved. With the intention of highlighting how different is the method described here, we present below a summary of all methods.

In this work, Green's function for the two-dimensional Helmholtz equation is obtained by directly solving this equation. First, a Fourier transform is used to reduce the two-dimensional problem into a one dimension problem which is relatively easy to solve. Next, by considering the corresponding inverse Fourier transform integral as a contour integral in the complex plane and performing a suitable change of the complex variable of integration, that integral can be considerably simplified \cite{DaviesTAM41}. It then becomes possible to identify this simpler integral with the forementioned Hankel function as well as to perform the calculations beyond equation (\ref{greenE160}).

The first already existing \emph{direct} method is given in reference \cite{ToscanoCoutoRBEF2}, where the calculation begins with the application of the two-dimensional Fourier transform. Then, in order to evaluate the inverse Fourier double integral, contour integration in the complex plane is used to perform one of the integrals, being necessary to carry out a detailed analysis to determine the correct prescription for circumventing the real poles of the integrand. Thereafter, the resulting integral, by means of a few manipulations, is converted into a known integral representation of that Hankel function, thus ending the calculation.

The other direct method can be found in references \cite{Duffy}, equations (5.1.14) to (5.1.16), or \cite{Economou}, section 1.2.2, where symmetry considerations are used to turn the problem one-dimensional, depending only on $\rho = |\vetor{r}-\vetor{r}\,'|$, in which a nonhomogeneous Bessel equation of order zero, exhibiting a delta function $\delta(\rho)$ on the right-hand side, has to be solved. The general solution of this differential equation is well known for $\rho\ne0$. Therefore, it only remains to determine the arbitrary constants; this is accomplished by imposing the correct conditions for $\rho\to0$ and $\rho\to\infty$ (to satisfy this condition at infinity -- the radiation condition -- it is easier to work with the general solution formed with the Hankel functions.)

One way of obtaining Green's function for the two-dimensional Helmholtz equation \emph{without solving directly} this differential equation is by employing the method of descent (see reference \cite{Courant-Hilbert}, Ch.~VI, \S12, 3), by means of which the solution of the two-dimensional problem is obtained by integrating the solution of the corresponding  easier three-dimensional problem with respect to the variable which spans the dimension being eliminated (the Cartesian variable $z$, in the case).

Another \emph{indirect} way (see reference \cite{Barton}, section~13.2.2) is as follows: one first relates Green's function for the wave equation, $G(\vetor{r},t\,|\,\vetor{r}\,',t')$, to Green's function for the Helmholtz equation, $\itGamma(\vetor{r}\,|\,\vetor{r}\,')$, and then, having determined the former (e.g., by descenting from the easier three-dimensional problem), he uses this relationship to calculate the latter. The first step is accomplished by noting that the stationary wave described by Helmholtz equation is a particular case of the generic wave motion described by the wave equation, from which it follows that $\itGamma(\vetor{r}\,|\,\vetor{r}\,') = \int_{-\infty}^{\,t } G(\vetor{r},t\,|\,\vetor{r}\,',t')\,e^{\iu\omega_0(t-t')}\,dt'$.

The importance of a new method for a problem already solved resides in the method itself, for it is likely to have other applications. This is true even if such method becomes more involved in that particular problem, because this may not happen in others. In fact, greater generality often requires more elaboration.

\begin{abstract}
{\bf Resumo}. Este trabalho descreve a aplicação de novas metodologias para o cálculo das transformadas de Fourier inversas que fornecem as funções de Green associadas às equações da onda e de Helmholtz em todo o domínio bidimensional.

{\bf Palavras-chave}. Equação da onda, bidimensional, função de Green.
\end{abstract}




\end{document}

